Author’s response to reviews

Title: The eCALM Trial: eTherapy for Cancer AppLying Mindfulness Online
Mindfulness-based cancer recovery program for underserved individuals living with cancer in Alberta: protocol development for a randomized wait-list controlled clinical trial.

Authors:

Kristin A Zernicke (k.zernicke@ucalgary.ca)
Tavis S Campbell Dr. (t.s.campbell@ucalgary.ca)
Michael Speca Dr. (Michael.Speca@albertahealthservices.ca)
Kelley McCabe-Ruff (kelley@emindful.com)
Steven Flowers (steve@mindfullivingprograms.com)
Dale A Dirkse (dadirkse@ucalgary.ca)
Linda E Carlson Dr. (lcarlso@ucalgary.ca)

Version: 2 Date: 11 November 2012

Author’s response to reviews:

Could not view cover letter materials following initial email confirmation of manuscript submission, and so the cover letter materials have been uploaded again as instructed in the confirmation email.